Author’s response to reviews

Title: Childhood and adolescent phenol and phthalate exposure and the age of menarche in Latina girls

Authors:

Alexandra Binder (amlbinder@gmail.com)
Camila Corvalan (ccorval@gmail.com)
Antonia Calafat (aic7@cdc.gov)
Xiaoyun Ye (xay5@cdc.gov)
Verónica Mericq (vmericq@med.uchile.cl)
Ana Pereira (anitapereiras@gmail.com)
Karin Michels (k.michels@ucla.edu)

Version: 2 Date: 12 Mar 2018

Author’s response to reviews:

Editors’ comments: The reviewers are happy with the revision, and so are we. Before we send your manuscript to production, we ask you to make some small modifications in accordance with our instructions to authors. Please remove the numbering of the sections, and also the subheads in the Results. We also recommend that you move last para in Results to the Discussion, perhaps as the first para.

-> We again thank the Editor and Reviewers for their thorough and constructive feedback. The authors have made the final recommended changes, including the removal of section numbering and removal of the Result sub-headers in accordance with the journal conventions. We have also integrated the last paragraph of the Results into the first paragraph of the Discussion, as suggested by the Editor.